



Article Urban Intersection Classification: A Comparative Analysis

Augusto Luis Ballardini ^{*,†}[®], Álvaro Hernández Saz [†][®], Sandra Carrasco Limeros [®], Javier Lorenzo , Ignacio Parra Alonso [®], Noelia Hernández Parra [®], Iván García Daza [®] and Miguel Ángel Sotelo [®]

> Computer Engineering Department, Universidad de Alcalá, 28805 Alcalá de Henares, Spain; alvaro.hernandezsaz@uah.es (Á.H.S.); sandra.carrascol@uah.es (S.C.L.); javier.lorenzod@uah.es (J.L.); ignacio.parra@uah.es (I.P.A.); noelia.hernandez@uah.es (N.H.P.); ivan.garciad@uah.es (I.G.D.); miguel.sotelo@uah.es (M.Á.S.)

* Correspondence: augusto.ballardini@uah.es

+ These authors contributed equally to this work.

Abstract: Understanding the scene in front of a vehicle is crucial for self-driving vehicles and Advanced Driver Assistance Systems, and in urban scenarios, intersection areas are one of the most critical, concentrating between 20% to 25% of road fatalities. This research presents a thorough investigation on the detection and classification of urban intersections as seen from onboard front-facing cameras. Different methodologies aimed at classifying intersection geometries have been assessed to provide a comprehensive evaluation of state-of-the-art techniques based on Deep Neural Network (DNN) approaches, including single-frame approaches and temporal integration schemes. A detailed analysis of most popular datasets previously used for the application together with a comparison with ad hoc recorded sequences revealed that the performances strongly depend on the field of view of the camera rather than other characteristics or temporal-integrating techniques. Due to the scarcity of training data, a new dataset is created by performing data augmentation from real-world data through a Generative Adversarial Network (GAN) to increase generalizability as well as to test the influence of data quality. Despite being in the relatively early stages, mainly due to the lack of intersection datasets oriented to the problem, an extensive experimental activity has been performed to analyze the individual performance of each proposed systems.

Keywords: intersection classification; scene understanding; self driving; intelligent transportation systems; CNN; GAN; RNN

1. Introduction

Estimating the scene in front of a vehicle is crucial for safe autonomous vehicle maneuvers, and it is also key to Advanced Driver Assistance Systems (ADAS). Even though performance and availability of scene understanding systems increased over the past years, technology seems to be far from the requirements of SAE (https://www.sae.org/ standards/content/j3016_202104, accessed date 16 August 2021) full automation level, particularly regarding urban areas and contexts without a strict Manhattan-style city plan, i.e., large building blocks surrounded by simple straight avenues. The transition towards full automation requires reliability under all circumstances, including ordinary middle and small cities where narrow and strangled roads are the most common. In 2019, approximately 22,700 people died throughout Europe due to traffic accidents according to the European Commission's latest report [1], and the last Annual Accident Report of 2018 [2] show us that the 19.4% of EU road fatalities comes from at-grade intersection areas. Similar results come from the reports of the United States National Highway Traffic Safety Administration (NHTSA), which show us that in the 2015–2019 period, at-grade intersections areas concentrate more than 40% of motor vehicle and 25% of fatal crashes [3]. Indeed, intersection areas are one of the most critical for both drivers and pedestrians. It is where vehicles' paths might cross at some point and where pedestrians are most likely to cross



Citation: Ballardini, A.L.; Hernández Saz, Á.; Carrasco Limeros, S.; Lorenzo, J.; Parra Alonso, I.; Hernández Parra, N; Garcia Daza, I.; Sotelo, M.Á. Urban Intersection Classification: A Comparative Analysis. *Sensors* **2021**, *21*, 6269. https://doi.org/10.3390/s21186269

Academic Editor: Francesco Bellotti

Received: 16 August 2021 Accepted: 12 September 2021 Published: 18 September 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the roadway. Consequently, within the navigation system of an autonomous vehicle, it can be considered of paramount concern to have a reliable system able to identify this specific traffic scene. For example, the vehicle control system might notice the number of road entries to the upcoming crossing and therefore estimate from which trajectories can come either other vehicle or pedestrian on a collision course. Besides the safety field, knowing the topology of the upcoming junction can allow the vehicle to perform safe maneuvers without the need to have a complete and detailed map of the area, as a human being would do when asking for directions in a city he does not know. In addition, we also believe that we can exploit the detection and classification of intersections for multiple purposes, such as an input to high-level classifiers of other driver's maneuvers. This might ease the prediction of position and intentions of Vulnerable Road Users (VRU) [4], or plan how to drive centered and safely without road markings. Toward this goal, existing intersection detectors were often paired with localization algorithms coupled with an external Global Navigation Satellite System (GNSS) and map data coming from providers like TomTom or HERE, which started to provide maps commonly referred to as High-Definition maps (HD maps). Even though the benefit of a priori knowledge on the understanding of the surrounding environment is undisputed, as it allows the system to narrow the scene reconstruction uncertainty, we also argue that relying only on updated maps might jeopardize the safety of autonomous driving systems themselves. Given the impact of vehicle crashes on human lives and the plethora of driving scenarios needed to consider, it follows that autonomous driving cars should be able to overcome, if not all, at least most of the common driving events that arise in actual traffic conditions. This work presents an in-depth study of the intersection classification problem and, following previous literature contributions, we defined seven basic geometries to which we will refer throughout the remainder of our contribution. Toward this goal, we explored the capabilities of state-of-the-art neural network approaches for image classification, considering different approaches such as single-frame, video analysis, and 3D-networks; non-traditional learning techniques for the specific intersection analysis subject; as well as different data source, e.g., RGB cameras and Light Detection And Rangings (LiDARs) sensors. The availability of intersection data is crucial in the Deep Neural Network (DNN) context, and both industries and research entities have worked hard during the last years to create large datasets explicitly conceived for autonomous driving research purposes. However, most of these datasets are collected in extensive metropolitan areas, in which ample avenues are the majority. On the one hand, tackling the classification problem in these scenarios is usually of extreme difficulty due to mere visibility issues, see Figure 1, that typically arise in these conditions. Intersections areas on these specific scenarios are generally well defined and thoroughly regulated with horizontal lane markings and traffic lights, circumstances that high-level self-driving systems can exploit for their advantage. This intuition was confirmed after a comprehensive yet intensive survey on the most popular self-driving datasets publicly available, including ApolloScape [5], Argoverse [6], Lyft-5 [7], nuScenes [8], and Pandaset [9]. On the other hand, intending to provide similar aid to the same high-level systems but in less structured environments, in this paper we delve into the identification of intersections in less metropolitanized areas. Note that well-known KITTI [10] and KITTI-360 [11] datasets, even though collected in the mid-sized city of Karlsruhe, Germany, contain an extremely low number of intersections, in the order of a few hundreds, including all the *City* and Residential sequences of the Raw-Data section. Moreover, the focal length of the cameras used in this dataset makes the recognition task harder, as, over short distances, it is not easy to see the whole intersection (whereas over long distances intersecting streets are unnoticeable). Last, this dataset contains only images with similar lighting conditions. For this reason, we decided to record a new dataset in the cities of Alcalá de Henares and Meco, Madrid, Spain. This dataset contains more than 530 intersections in different light conditions, more than all of those contained in the overall KITTI and KITTI-360 sequences. Concerning the camera configuration, we opted for a consumer action camera mounted on different vehicles as visible in Figure 2 that provided excellent visibility of

the surrounding area, good lighting handling, and easy operation. To further increase the number of intersections during the training phase, we also exploited the generative capabilities of state-of-the-art Generative Adversarial Networks (GANs) [12] architectures, introducing new photorealistic images of intersections. Extensive experimental evaluations show the pros and cons of each of the different investigated techniques. The remainder of this work is organized as follows. Section 2 provides an overview of the existing literature on intersection classification. Section 3 presents the datasets used in this work along with technical details that highlight why we proposed a new intersection dataset. Section 4 introduces the techniques proposed in this work, and Section 5 presents the experimental activity and the ablation study that allowed us to analyze the performances of the proposed techniques. Finally, in Section 6, we present our conclusions and the future works.





Figure 1. Typical intersections in the well-known ApolloScape (**a**,**b**), Lyft-5 (**c**,**d**), NuScenes (**e**,**f**), and Pandaset (**g**,**h**) datasets. The intersection geometry configuration on these Manhattan-like environments is simple and repetitive, and the dominant difference consists only on the traffic conditions. Moreover, it is of extreme difficulty to generalize these kinds of settings to mid-sized cities.



Figure 2. Our vehicle with the attached consumer action camera.

2. Related Work

The relevance of road intersection detection can be noticed from the interest in the problem coming from different research communities and traffic regulation agencies [13,14]. From a technical perspective, we can first distinguish approaches that exploit images from stereo or monocular camera suites, algorithms that only rely on LiDAR sensors, or finally, a combination of the previous. One of the earliest works in this research field can be found in [15], where two standard computer vision methods are presented. The authors proposed a first method to extract road boundaries from the images, matching them to a predefined intersection model using heuristic methods, followed by a second system aimed at finding free space corridors through which it is safe to drive. Later, in 2013, Andreas

Geiger published his Ph.D. thesis [16] in the field of urban scene understanding. In his thesis, he proposed two new approaches designed for detecting and classifying intersection areas that, in many cases, can be considered the starting point of the recent research on this field. The first approach consists of a new intersection modeling system with flexibility in the number of intersecting streets and crossing branches' location, orientation, and width. The second proposal involves an efficient learning and inference algorithm for scene understanding, based on Markov Chain Monte Carlo Sampling and belief propagation. In different contexts but with a similar approach, the authors of [17,18] exploited RGB images from vehicle front-facing cameras and standard computer vision techniques to create temporally integrated occupancy grids that were in turn compared to predetermined shapes to assess the presence of upcoming intersections. The approach proposed in [19] is substantially different. Here, the input is solely based on point clouds obtained by LiDAR sensors. These raw data go through three independent processes that generate a binary classification of whether they belong to an intersection or not. The first process is aimed to extract the most relevant features from the point clouds. These features capture appropriate geometric and statistical properties that are consequently classified through different classifiers, such as Multilayer Perceptron (MLP), Support Vector Machine (SVM), and AdaBoost. Finally, as these methodologies produce a single frame classification, the third step involves Hidden Markov Model (HMM) and Conditional Random Field (CRF) approaches to model the observation time series. Despite working with different kinds of sensors, i.e., LiDAR instead of cameras, the work presented in [20] is closer to our research goals and uses Convolutional Neural Networks (CNNs) approaches coupled with transfer learning techniques to classify intersections. Starting from a Fully Convolutional Neural Network (FCNN) trained to predict the future path of the ego-vehicle, the authors proposed an architecture called Intersection Net that exploits the encoder's features to perform the intersection classification in terms of six basic classes, including 3-way, 4-way, and roundabouts. So far, the presented research based on deep learning has focused on detecting or classifying intersections once they have been reached. However, it has been seen in [19] that temporal integration can be an asset to be considered. In [21], this research line is further investigated and a Long Short-Term Memory (LSTM) [22] architecture is attached to a CNN to solve the "Have I reached the intersection?" problem. The difference between LSTMs and other types of architectures is that they can perform temporal integration thanks to a "short-term memory" and a "long-term memory" that allows them to remember what they have previously processed to execute the current prediction better. The workflow of their final architecture is as follows. First, a sequence of ordered images is supplied to the CNN, which will in turn return a sequence of feature vectors. These vectors are packed and sent to a LSTM network, which integrates them temporarily to extract information to binary classify the sequence as "intersection" or "not intersection". From a technical perspective, our work extends theirs in the following ways. First, our work is aimed not only towards a binary classification task, but, following previous state-of-the-art approaches, we identified a broader set of intersection classes, even larger than their final three-class expansion. We also deep investigated different recurrent network architectures as well as fusion layers on a much larger set of data, to systematically evaluate the performance given by each different component. Last, it is important to notice that the authors highlight the lack of an existing dataset specifically designed and publicly available for evaluating the performance of an intersection detection method. Together with the KITTI dataset, our proposal address this issue, as we believe that much more data and research is needed in this field. The work presented in [23] maintains a similar approach about temporal integration, but instead of detecting whether the intersection is reached, it tries to classify them into seven basic typologies. For this purpose, their system uses two data sources that the authors called Input-F and Input-T. Input-T is an RGB image taken just before reaching the intersection, and Input-F is a sequence of images taken while crossing the intersection. The authors used an architecture composed of three different modules, F-Net, T-Net, and I-Net to process these two data sources. In detail, T-Net is a VGG16 network in which the classifier is replaced by two fully connected layers trained to distinguish between the seven classes, freezing the other weights. F-Net instead is a combination of Inception-V3 and an LSTM architecture. The goal of Inception-V3 is to extract the optical flow from the image sequence so that the LSTM can classify them between straight crossings, right-facing crossings, and left-facing crossings. Finally, I-Net is the architecture in charge of integrating the previous two architectures' results to return a final classification using Bayesian multimodal information fusion [24]. Conceptually, some of the approaches are also investigated in our work, i.e., the use of LSTMs or VGG-based systems. However, the remainder of our work shows how better performances can be achieved by using simpler system configurations that do not require multi-level architectures to perform the intersection classification.

2.1. LSTM

Long Short-Term Memory networks are a type of Recurrent Neural Network (RNN) developed by Hochreiter et al. [22] to avoid vanishing gradient problems that are intrinsic in the vanilla RNN. This architecture was initially conceived for Natural Language Understanding and time series prediction tasks. However, in the last decade, they have been introduced in the computer vision world in combination with convolutional architectures. In [25], the authors proposed this combination to deal with the problem of image captioning (i.e., generating sentences describing an input image). Following this work, new models emerged adopting this combination of LSTM and CNN networks to the spatio-temporal domain of video sequences. In [26], a model called Long-term Recurrent Convolutional Networks (LRCN) is presented. This model combines a CNN backbone for spatial feature extraction from raw video frames and an LSTM network for temporal feature extraction. By summarizing the features obtained with the recurrent network, a final action is inferred for the whole input sequence. While this model is initially applied to human action recognition, in [21] this combination of networks is applied to binary intersection detection (i.e., probability of an intersection in the input video). A similar model is used in [27], but combining optical flow information with the raw video sequence, improving state-of-the-art results in several action benchmarks. As LSTM networks are not suitable for high-dimensional data due to their fully-connected nature, in [28] ConvLSTM was proposed. In this model, LSTM input-to-state and state-to-state transitions are replaced by convolutional architectures, making it suitable for image data. This model has been applied to several tasks such as medical image segmentation, video frame interpolation, and super-resolution. However, the LSTM standard model continues to be the standard among recurrent models and video action recognition is currently dominated by CNN 3D models.

2.2. GAN

It is a widespread practice to augment image-based datasets in ways that are appropriate for a given dataset. In general, data augmentation helps to stop a model from overfitting to the data and allows users to make small datasets many times larger. However, a sufficiently powerful classifier often still overfit to the original examples, which is why GANs are particularly useful here. GANs are well known for their utility in generating fake data when real data is either too hard to come by or expensive to acquire. One of the benefits of using GAN data augmentation is that they are often better than handcrafted synthetic examples. Another pro of GAN data augmentation is that it can generate labeled examples using a conditional setup, which is really useful in case of imbalanced datasets. Finally, it has been shown to improve downstream model generalization across a variety of tasks, including image segmentation, classification, and detection [29–34]. Several approaches have been proposed for image synthesis with generative adversarial networks. In our specific application, data augmentation can enable better generalization over different environments or weather and lighting conditions, and it can also be used to balance our dataset or for sensor correcting, such as fixing noisy inputs and sensor modeling. The synthesis can be done in two- or three-dimensional spaces, in addition to spatio-temporal spaces

such as videos. For example, Point Cloud GAN proposed a twofold modification to GAN for learning to generate point clouds [35]. Moreover, they provide a study for transforming images into point clouds. PrGAN [36] generates a distribution over 3D structures given 2D views of multiple objects taken from unknown viewpoints. On the other hand, Temporal GAN (TGAN) [37] generates videos using a temporal generator and an image generator learning a semantic representation of unlabeled videos. However, best results are still found in 2D Synthesis with state-of-the-art models such as StyleGAN [38], StyleGAN2 [39], and SWAGAN [40], which incorporate wavelets throughout its generator and discriminator architecture enforcing a frequency-aware latent representation, or StyleGAN extension for training with limited data using Adaptive Discriminator Augmentation (ADA) [41].

3. Dataset

We addressed the intersection classification challenge by trying to exploit the most popular dataset available in the autonomous driving field, even though not explicitly designed for intersection detection purposes. Using these datasets would help us to easily compare our results from some of the previous works mentioned in Section 2. However, the first issue we faced was related to the locations where these datasets have been recorded. As can be seen in Figure 1, these datasets were collected inside huge metropolitan areas like San Francisco (Pandaset/Lyft5); Beijing; and other cities in China (ApolloScape/Baidu), Boston, and Singapore (nuScenes). The intersection visibility in these conditions is hampered due to several reasons: the width of the avenue and the related lens' field of view, the separation of travel directions that make two-way traffic streets look like one-way ones, and the massive number of vehicles. One possible solution for the identification and classification in these extreme conditions would be, for example, exploiting the detection of traffic lights and the analysis of other car trajectories, such as in [16]. On the other hand, less urbanized areas like the ones in Figure 3 can benefit from the capabilities of classification networks. To the best of our knowledge, the only public dataset previously considered for the intersection classification problem in mid-sized cities is the KITTI dataset and its recent KITTI-360 extension. There are two issues that, after careful analysis, prevent state-of-the-art approaches based on image-based detectors from obtaining excellent results. First, as can be seen in Table 1, the number of intersection frames is relatively low. This makes the training phase of any network extremely challenging.





(b)





Figure 3. Some examples from the mid-sized city of Karlsruhe for the KITTI and KITTI-360 datasets and Alcalá de Henares (our dataset). The reader can appreciate the differences in the spatial distribution of street elements with respect to Figure 1, as well as the change in the field-of view between the KITTI cameras (a-c) and our dataset (d-f).

Besides the frame availability, another interesting number to analyze is the actual number of intersections. As can be noticed in Figure 4, and considering only the numbers associated with KITTI-ROAD sequences (a subset of the original KITTI dataset), the number of per-type intersections is critical, far below the expected number of elements usually typically used for DNN training's, especially for those networks that use sequences instead of the mere number of frames. Note that, for a given intersection, the visual appearance of the scene on consecutive frames has minimal changes. Together with the low frame availability, this forced us to pay special attention to the common dataset split phase: randomly choosing frames from the whole dataset was not an option, due to the multiple frames associated with every intersection. By randomly selecting frames, it would have been possible to include similar frames of the same intersection into both training and validation or testing, frustrating the separation efforts. As there is no easy solution for the availability issue, these considerations led us to create a much more extended dataset of intersections recorded in the surrounding area of Alcalá de Henares, Madrid, Spain, during the first half of 2021. Together with a set of scripts and additional descriptions, all the images will be released to the community and publicly available on http://invett. es/intersectiondataset. A current subset of the dataset, which contains all intersection geometries in different weather conditions and seasons, is used in this work, and a few images can be appreciated in Figure 5.

Table 1. Intersections frames per-class, all datasets.

Sequence	0	1	2	3	4	5	6
Alcalá-1 (jan26)	4693	1321	1467	2144	4429	2291	5374
Alcalá-2 (feb12)	2686	745	446	1893	1973	1287	2948
KITTI-ROAD	X	308	21	452	52	523	1035
KITTI-360	1190	711	999	1040	706	1048	1129
Total	8569	3085	2933	5529	7160	5149	10,486

Number of frames associated with each intersection class, for each of the used datasets. Please note that (i) in the KITTI-ROAD sequences, we did not add type-zero intersection frames, i.e., straight roads, to keep the balance between classes; furthermore, as our intentions are more focused on the understanding of real crossing context, this special kind of intersection has low relevance. (ii) The number of frames is not related at all with the number of different intersections, as multiple consecutive frames can be associated with the same area. Besides being recorded from a different position, we do not perform the same trajectory control procedures to avoid exact matching frames.



Figure 4. Distribution of intersection types by the seven considered road configurations (0–6) (**a**) and sequence length (**b**) for KITTI, KITTI-360, and the two sequences of our Alcalá de Henares dataset. As the reader may notice, the number of intersections introduced with our sequences doubled the number of intersections for every class. Moreover, the average frames associated with each of the intersections is also increased. KITTI is recorded at 10 Hz, while our sequences are recorded at 30 Hz.

Among the datasets used in our work, one substantial difference lies in the availability of *single* vs. *stereo* camera head and *camera* (single-sensor) vs. *camera and LiDAR* (multi-sensor) configurations. This difference allowed or denied some of the configurations explained in Section 4.4.2.



Figure 5. Each column shows three examples of the seven configuration geometries that were used in this work (with the *so-called* canonical images shown in the upper part of each column). The RGB examples come from our recorded dataset. We recorded the images at different times of the day using a different camera setup for more than 3:30 h of recording time, including more than 500 intersections.

4. Technical Approach

This study's purpose consists of the assessment of different methodologies addressed to detect the geometry of an upcoming intersection outside metropolitan areas. Toward this goal, we have evaluated the classification capabilities of state-of-the-art DNNs systems, including networks aimed at analyzing single-frame images as well as video sequences. Moreover, concerning the single-frame image analysis, we wanted to verify the performance gain that the metric learning and the Teacher/Student learning paradigm have with respect to a standard classification baseline. Different loss functions, including triplet-loss techniques, were also investigated. Each of these network configuration and training methods is carefully described in the following subsections, which explore our extensive research activity in detail. For the sake of simplicity, Table 2 and Figure 6 summarize all the presented approaches.

Table 2. Overall scheme of training approaches.

Learning Scheme	Key Features
Baseline	End-to-End Classification Standard RGB image classification method
Metric Learning	Learning to classify by using distances between embedding vectors
Teacher/Student	Two networks are sequentially trained metric learning can also be applied
LSTM	These networks are well suited to process time series data
PyTorchVideo	Deep Learning library for video understanding research



Figure 6. The picture includes all the approaches presented in this work, including Direct per-frame classification (Section 4.1), Teacher/Student (Section 4.4), Metric Learning (Section 4.2), and Multi-frame analysis (Section 4.5), together with the GAN-based data augmentation scheme. As the reader can notice, all the approaches except the direct per-frame classification involve using embedding vectors obtained from DNN systems, which in turn are further processed using the techniques described in the remainder of this work. The clock symbol indicates when a time-series analysis is involved.

4.1. RGB Baseline

Our preliminary work [42] started our evaluations by assessing the end-to-end classification capabilities of two well-consolidated networks for image classification: ResNet [43] and VGG [44] networks. Throughout the subsequent research here presented, more network architectures have been incorporated to explore the capabilities of other backbones more intensively. In addition to all the common ResNet and VGG implementations, MobileNet-V3 [45] and Inception-V3 [46] architectures have also been investigated. The performances of each of the networks, as well as the evaluated loss functions, will be discussed in Section 5. Besides representing the more trivial learning paradigm, these end-to-end configurations allowed us to create an architecture baseline to compare each additional system. Following our first preliminary work in [42], regarding the KITTI and KITTI-360 sequences, we used the RGB images from the left camera of the stereo rig. For all the sequences of either KITTI or our datasets, in a similar way to what was proposed by other authors, we also prepared a second set of images containing 2D-homographies of the original images. Our intent is to create Bird's Eye View (BEV) images for each of the correspondent RGBs, trying to help the classification network by eliminating cluttering elements of the scene, e.g., above horizon elements, while at the same time emphasizing the intersection geometry. Together with the images directly acquired from the camera, these two sets of images allowed us to perform a first classification assessment of the networks mentioned above and evaluate the improvements described in the following sections.

4.2. Metric Learning

The so-called *metric learning* represents an exciting and emerging technique in the machine learning community. This notation, used as a generic term to indicate a distance, a correspondence, or difference between elements, allows for a slightly different classification approach from the one presented in the previous section. As it is well described in [47], with a metric-learning technique, we can shift the classification problem towards an optimization problem so that pairwise or triplet-based distance constraints between items are both enforced and minimized using a specific loss function. From a technical perspective, the distance is evaluated starting from the *embedding vectors* associated with a couple of input images, where the embeddings are generated by a function $f(\cdot)$ in the form of DNN, see Figure 7. The idea is that given three images and the associated embedding vectors called *anchor (a)*, *positive (p)*, and *negative (n)*, where the anchor is the query point in the vector space, a generic distance *d* with respect to the positive and negative points in the vector space satisfies the Equation (1).

$$d(f(a), f(p)) < d(f(a), f(n)) \tag{1}$$

In this work, we used two different implementations of the metric-learning technique depending on the specific problem we were analyzing, including an ad hoc implementation and the much more exhaustive work presented by Kevin Musgrave in [48] (please note that in the remainder of the manuscript we refer to this work as the metric library). This package allowed us to evaluate different distance functions apart from the generic L^2 distance, e.g., Cosine-Similarity and SNR-Distance [49]. Moreover, we used the concepts of *Triplet Margin Miner* to use specific positive-anchor and negative-anchor pairs. These concepts will become more apparent after the next section.



Figure 7. The distance between the two images is calculated using the numerical N-dimensional embeddings resulting from a generic DNN network. For example, considering the ResNet model, the resulting embedding E_x dimension is 512. The intuition is that for different intersections the distance d' between embeddings is larger than the distance of same-class intersection d''.

Triplet Scheme

Within the plethora of metric-learning different patterns, we used the *triplet* approach described in [50], where a set of three images (M_1^a, M_2^s, M_3^d) belonging to the same domain, i.e., M, composed of one *anchor* class image M_i^A , a *same* class sample (positive-anchor) M_j^S , and a *different* class sample (negative-anchor) M_k^D , are passed through the triplet margin loss function. Again, the idea is to have a DNN generating a high-dimensional embedding

vector associated with input images. The distance between vectors belonging to different intersection classes is higher than the distance derived from the vectors of same-class intersections. This loss function is defined similarly to each of the two parts of Equation (1), as follows:

$$\mathcal{L} = \sum_{i} [d(f(M_i^A), f(M_j^S)) - d(f(M_i^A), f(M_k^D)) + m]_+$$
(2)

where $[\cdot]_+$ means $max(0, [\cdot])$ and $d(x_i, y_i) = ||x_i - y_i||_p$ with p as the norm degree for pairwise distance and m is an extra scalar value used to extend the *margin* between the embeddings. As the reader might guess, not all the positive and negative triplets will have similar scores. Unexpected geometries might also generate confusion and lead to unsatisfactory results. For this reason, we proposed using the *Triplet Margin Miner* feature provided in [48], which allows us to specify a minimum distance threshold for the generation of the triplets.

4.3. RGB Metric

We began our investigation by directly verifying the benefits introduced by the metric learning approach. First, according to Equation (2), we evaluated different distance functions. Starting from the basic $\|\cdot\|_1$ and $\|\cdot\|_2$, we have expanded the distance function set to assess the impact of Signal-to-Noise Ratio and Cosine Similarity functions provided within the metric-library. As mentioned, Section 4.2, the metric library also offers the opportunity of using a *miner*. This solution allows us to collect those pairs or triples that exceed a preset threshold before calculating the loss function itself, working as a difficulty discriminator between the elements. Specifically, the Triplet Margin Miner has been used for this work, with two of its possible configurations: "all" and "hard". The first of these configurations lets us calculate the loss value by using all the triplets that violate the preset margin. The second one allows for creating of a subset of the previous triplets, where the positive example is further away from the anchor than the negative example. These two configurations allow us to focus our training efforts on the most challenging examples for the neural network. However, for the sake of completeness, we also performed training without any miner, letting the system use any possible triplet regardless of its difficulty. As this training methodology works directly with distances, it is necessary to identify a metric to evaluate the performance of the system. The metric library provides different functions that can be used in this situation, such as Mean Average Precision@R (MAPR), Mean Average Precision (MAP), Precision-at-1 (PA1), and R-Precision (RP). All these values have been recorded during the training process to assess how the training was running. However, the value is taken into account to run the validation patience and select the best training is MAPR, as this metric operates directly on the embedding space and provides better information [51]. Once the model has been trained, it is necessary to find a way to use it, as an embedding does not directly refer to a specific intersection type. An easy and apparently trivial way could be to use the metrics above to calculate the minimum distance to a specific class representative vector. For example, using a clustering procedure, the representative vector might be the centroid of each cluster. However, apart from relying on such elements as centroids, we also evaluated the following methods. Following the distance concept, a first approach consists in using the Mahalanobis distance rather than the Euclidean distance to calculate the classification of each new example in such a way to capture better the distance with respect to the learned concepts. The second method consists of a transfer learning adaptation: once the network has been trained using the previous proposals, its weights are frozen, and a fully connected layer is trained to classify the embeddings returned by the network. Finally, the third proposal consists of training an SVM classifier using the embeddings of the training and validation sets. Once the classifier is trained, the implementation of the system in a real environment would consist of two steps: obtaining the embedding from the trained architecture and its classification employing the SVM.

All the training activities performed with this methodology, in addition to the three datasets already explained, have been carried out also with different data transformations which will be explained in detail in Section 4.4.2.

4.4. Teacher/Student and the Intersection Model

The second learning paradigm we wanted to evaluate with respect to the standard end-to-end scheme was the so-called *Teacher/Student*. Among all the possible applications, the idea behind this paradigm includes transferring knowledge from a simple domain to a much more complex one.

4.4.1. The Intersection Model

From a technical perspective, the starting domain from which we propose to transfer the knowledge consists of a synthetic set of simple Bird's Eye View (BEV) images generated with the intersection generator proposed in [18,52]. This simple intersection generator, shown in Figure 8, generates all seven configuration classes commonly used in the literature in the form of binary masks, such as those shown in Figure 9. We refer to these images as Model-Based Bird's Eye Views (MBEVs). These masks contain the shape of the most common intersections entities that can be found in mid-sized cities. The model parameterization allows us to generate different intersection geometries, i.e., changing the number and position of intersecting roads, the center position with respect to the generated image, and, finally, the carriageway of each road segment involved. Without pretending to be the definite intersection generator, as many configurations still cannot be represented with our proposals, e.g., complex intersections that can be typically seen in big metropolitan areas, this model allowed us to assess the intersection classification capabilities of the evaluated network configurations using the datasets proposed in Section 3. This model acted itself as a trivial data augmentation scheme for the CNN during the training phases of our teacher networks. To mimic the way in which some of the training images are created, we optionally added an increasing amount of random noise starting from the bottom part of the mask in a line-by-line fashion, as visible in Figure 8. More details will be provided in the following sessions.



Figure 8. On the top of the image, the seven so-called *canonical* intersection classes along with the model used to generate the training dataset. In the leftmost part of the figure, identified with the label "*w/o noise*", a triplet consisting of two samples of the canonical *type-three* (for simplicity also shown in the last box of the row) and a different one, e.g., *type-five*. Before passing these images to the teacher network during the training phase, random noise is added to each image by increasing extent along the vertical direction. The effect is shown in the rightmost part of the figure identified with the label "*w-noise*".



Figure 9. More examples of MBEV images generated with the model in Figure 8.

4.4.2. RGB Pre-Processing

To exploit the MBEVs images generated with the intersection model, we created a set of three different pipelines to transform the RGB images into a similar viewpoint. This set includes the following.

- 3D-Generated Bird's Eye Views (3D-BEVs): this first transformation, applicable only to datasets that provides stereo camera configurations, creates a BEV representation of the scene using a 3D-reconstruction process; it is the most accurate 2D plan view that can be generated from images, as no distortions are introduced in this procedure. The effect of the virtual camera can be seen in the center box of Figure 10. We used the work in [53] to generate the depth image that, in turn, allowed us to create a 3D representation and then the desired 2D image. Please note that having a 3D representation allows us to change the virtual camera position retaining the scene's consistency and simultaneously acting as a data augmentation methodology.
- Masked 3D-Generated Bird's Eye Views (3DMASKED-BEVs): for this second transformation, we extended the previous pipeline by including the results in [54], to remove the 3D points that do not belong to the road surface and thus generate an image containing only road pixels. The main insight here is to evaluate whether the classification may benefit from less cluttered yet pre-segmented images. For this purpose, we combined the previous depth-image to the generated road-mask before creating the 2D view. The downside with this variation is that the approach in [54] only works with LiDAR data.
- Warping with Homographies (WARPINGs): this last transform tries to overcome the limitation of stereo and LiDARs availability at the cost of introducing distortions in the generated images. We applied standard computer vision techniques to create a homography between the RGB image and the desired 2D image. As homographies are only defined for flat surfaces, but common roads do always have at least slight deformations, this image transformation introduces distortions to the final image, as depicted in the *warping* images in Figure 10. Moreover, as we used fixed homographies, the actual attitude of the vehicle introduces similar distortion effects as the vehicle moves along its route. Nevertheless, as we have more than one frame-per-intersections in our videos, we can conceive this effect as a data augmentation scheme, as the same intersection visually appears different across successive frames. Please notice that this scheme is equivalent to the one proposed for the RGB-Baseline described in Section 4.1.

In addition to the basic RGB data, the three configurations were evaluated with respect to the learning schemes presented in our work. Please notice that 3D images are created only when LiDAR data is available.



Figure 10. The figure depicts the overall pipeline used to generate the 2D planar images used in this work (apart from the original RGB-left camera). Among them, only the 3DMASKED-BEV uses the LiDAR data. The central part of the image shows the two methods of our *Data augmentation scheme*, i.e., the 3D-based realistic virtual camera that exploits the 3D point cloud-based reconstruction of the image (from AANET and 3D-Deep networks) and the 2D-based homography that usually generates substantial image distortions.

4.4.3. Applying the Teacher–Student Paradigm

The central insight besides the Teacher/Student paradigm is to learn a shared embedding space between the 2D images created with the intersection model and those generated by the aforementioned transformation pipelines. This approach is inspired by the work of Cattaneo et al. [55], which performs visual localization using 2D and 3D inputs in a bi-directional mode, teaching two networks to create a shared embedding space and thus enabling a two-way localization, starting either from 2D or 3D inputs. Recalling the metric technique described in the first part of Section 4.2, the teacher-student paradigm introduced some minor changes, in particular to Equation (1). In detail, given two instances of the same intersection class belonging to different domains, e.g., Class 0 in domains D1 and D2 $(D_{C=0}^1 \text{ and } D_{C=0}^2)$ such as D1 is the space of the images with our intersection model and D2 is the RGB-transformed space, and two different non-linear functions $f(\cdot)$ and $g(\cdot)$ represented in the form of DNNs, the distance between the embeddings is lower than any other negative intersection instance, e.g., $D_{c=2}^2$. Formally, given the *Intersection-Model* domain M and the *Camera* domain C such that $M = C = \{0, 1, 2, 3, 4, 5, 6\}$, each of which contains the seven intersection typologies considered in our intersection model, and given one element $m_i \in M$, then Equation (3) is satisfied $\forall i, j \in C | i \neq j$, where $d(\cdot)$ is a distance function.

$$d(f(m_i), g(c_i)) < d(f(m_i), g(c_i))$$

$$(3)$$

Regarding the teacher network, which is the first part to be trained, we trained it similarly as we have previously done with the RGB images, but this time we used the images generated from the intersection model, see Figure 11. Once the teacher model has been trained, we trained the student network using the preprocessed RGB images presented in Section 4.4.2 as input data in a way to obtain a similar embedding vector. Towards this goal, the loss-function of the student network is composed as follows:

$$\mathcal{L} = \sum_{i} [d(f(M_i^A), g(C_i^S))]$$
(4)

where *M* and *C* are the model-domain and camera-domain, respectively, and Mean Squared Error (MSE) was used as distance function $d(\cdot)$ between the embeddings. Differently from Equation (2), here the triplet scheme is replaced by a pairwise distance between the embedding vectors. Note that to maintain a consistent distance within same-class classifications, M_i^A elements were chosen not from the list of embedding vectors used in the training phase of the teacher network but rather from the average of 1000 new random samples generated after the teacher network was trained, i.e., never seen before from the DNNs. These per-class averages, i.e., cluster centroids, are shown in Figure 12 with black crosses and represent therefore our M_i^A set.

4.4.4. Training Details

A data augmentation process was introduced in both networks to avoid overfitting during the network's training phase. We generated a set of 1000 per-class intersection configurations by sampling from our generative model for what concerns the teacher network. We applied a normal random noise to the seven *canonical* intersection configurations on each parameter involved in the intersection generation, e.g., width, angle, and intersection center, in a measure of 2.0 m, 0.4 rad, 9.0 m, respectively. For what concerns the noise, starting from the bottom of the image, we added an increasing number of random noises to each line to mimic the 3D density effect of 3DMASKED-BEVs. Regarding the student network, as the low number of intersections present in the two KITTI datasets in comparison with the overall number of frames, we performed data augmentation adding a 6-DoF displacement to a looking-down virtual camera initially set at 10.0 m and 22.5 m above the road surface and at 17.0 m and 22.0 m in front of the vehicle for the KITTI and KITTI-360 respectively. Due to the nature of type-1 and type-2 intersection classes, which contain any kind of curve without a specific curvature threshold, we zeroed the rotation

along the vertical axis to limit the chance of assimilating these samples to the type-0 class. Our code leverages the PyTorch 1.6 learning framework [56], and both teacher and student images were scaled to images with size 224×224 pixels. At this time, despite its triviality, note that the point-density of 3DMASKED-BEVs is not constant over the distance with respect to the vehicle. Therefore, to simulate comparable MBEVs, we added a random noise proportional to the distance, see Figure 8.



Figure 11. The picture depicts the position of MBEVs in the vector space together with their pairwise distances (colored edges), visually showing the intuition behind the metric learning approach. Same class (green) and inconsistent, different-class (red) edges in panels (**a**,**b**), respectively, are used to find a metric in a way that similar images are grouped together in clusters, as visible in (**c**). The images shown here are generated with the intersection model presented in Section 4.4. Besides this first intuition of constraints in terms of *pairs*, we can imagine *triplets* to further increase the complexity of the optimization problem, helping the system to obtain better and wider distances between clusters.



Figure 12. The embedding space is visually represented using 7k samples from the intersection model, using T-Distributed Stochastic Neighbor Embedding (t-SNE) function. Apart from the few outliers, the image depicts how the Teacher Network can separate the seven classes generated with our intersection model. In black, we conceptually represent the centroid of each of the clusters.

4.5. Multi-Frame Analysis or Schemes

We continued our research by assessing the results of multi-frame learning schemes. As the system is intended for autonomous vehicles, it seems logical to think that the vehicle will get more information about the surrounding environment as it approaches the intersection, and the interpretation accuracies will increase as confirmed in [18]. Therefore,

implementing a system that can exploit the spatio-temporal information from a sequence of images until the vehicle is inside the intersection should give even better results.

4.5.1. Recurrent Neural Networks

With the idea of further refining the work done so far with the CNN networks, we investigated possible architectures that could be combined with them to perform temporal integration of the data. It seems logical to think that the system will be able to process more information and improve the results by temporal integration of the data. The two predominant models for this in state-of-the-art are the Gated Recurrent Unit (GRU) and LSTM architectures. These types of architectures typically have input packets of data vectors grouped into temporal sequences. As the CNN networks that have been used so far return embeddings, the only process necessary to group the two architectures lies in the temporal aggregation of the embeddings. For this, the following process has been followed. First, the images have been grouped into sets belonging to the same junction forming a single temporal sequence. These samples are then passed through an already trained CNN, which returns a sequence of embeddings. Each recurrent network was trained with the best-performing CNN previously trained for each dataset and data type configuration. Once the CNN has processed all the batch sequences, the shortest ones are zero-padded so that all the sequences that compose the data block supplied to the recurrent network have the same length. Finally, as both GRU and LSTM return feature vectors, it is necessary to classify them. Similarly, as has been done for the metric approaches, we used two different methodologies: the inclusion of a fully connected layer trained simultaneously with the recurrent architecture and the use of an SVM classifier in a similar way as described in Section 4.3. With both architectures, we made several architecture variations to perform an adequate ablation study. These variations range from the number of layers in the network or the hidden layer's size to the internal dropout of the architecture. The training has been performed using all the available datasets and respective transformations presented in Section 4.4.2. More detailed information on training will be provided later in Section 5.4.

4.5.2. Video Classification Networks

Apart from the recurrent architectures, a second exciting network design for temporal sequence analysis is represented by the expansion of standard 2D architectures. The basic idea is to feed a standard classification network with a set of consecutive images, allowing for the creation of so-called spatio-temporal filters. The main issue with this architecture is the huge increase in terms of network parameters with respect to the corresponding 2D network version, which makes the training phase harder and computationally demanding. Moreover, the 3D extension also seems to neutralize the benefits of pretrained models [57]. Extensions to these basic video recognition architectures were introduced in the very last months by the works presented in [58], where a family of efficient video networks is presented. A preliminary remark that must be made is that all of these architectures are usually employed for the detection of specific activities in video sequences. Typical datasets include KINETICS [59], Charades [60], EPIC-KITCHENS [61], or something-something [62], where a huge set of activities is provided in terms of short videos. However, it is our opinion that similar actions or high-level *concepts* could also appear in a sequence representing a vehicle approaching a type of intersection. With the idea of having a similarity between the action "picking something up" and "approaching a type-x intersection", and the promising results of these approaches, we decide to perform a set of preliminary experiments using the works proposed in [63] and generally contained within the PyTorchVideo framework. To the best of our knowledge, this is the first time this kind of approach is evaluated in the intersection classification context.

4.6. Artificial Data-Augmentation: GAN

As previously stated, one of the major issues that affects all the aforementioned techniques is the limited availability of intersections, primarily in terms of intersection instances rather than with regard to the number of frames. In an attempt to overcome this limitation, we implemented and compared several GANs frameworks, from more simplistic approaches like DCGAN, WGAN, and CGAN to the last state-of-the-art networks, such as StyleGAN2, SWAGAN, or StyleGAN-ADA, which includes an adaptive augmentation procedure to deal with limited data regimes [41]. Given the limited amount of data, the best results in terms of quality and regularization metrics were founded for SWAGAN and StyleGAN2 with the adaptive discriminator augmentation. SWAGAN [40] noticeably increases computational performance; however, some checkboard artifacts appeared in the images, and it achieved worse perceptual path length regularization. Therefore, we finally opt for StyleGAN2-ADA. Figure 13 depicts some generated examples of the latter after 800k epochs.



Figure 13. On the left, an example of the performances of StyleGAN-2. On the right, an example of the output of the StyleGAN-2-ADA network after 800k epochs. The number of face images in the FFHQ [38] dataset is impressively high in comparison with the few sequences of our combined sequences. For this training, we used all the images available in both Alcalá and KITTI sequences, without any split. However, to avoid presenting too similar images, we decimated the input images to have a constant frame rate equal to 1 fps.

Following the intuition that feeding the network with warped images could improve the generative process, as the network would be able to focus only on intersections, we performed different tests training the GAN with RGB images and with warped images. In order to balance less represented classes, we trained the GAN following a conditional setting. However, for RGB images, the best results were found for the non-conditional framework. Therefore, we create the new augmented dataset generating 10,000 random images and labeled them frame by frame, discarding those which do not depict a clear intersection. Figure 14 shows some examples of good and bad generated images. One issue we found with the generated images is that most of them were similar to the original ones. This is probably because the number of training images obtained with all the sequences of both KITTI and Alcalá dataset is more than six times smaller than the 30k training images used in the StyleGAN-ADA original contribution [41]. Notwithstanding, results show better classification performances with the augmented dataset.



Figure 14. Some examples of good generated RGB intersections, included in the new augmented dataset (**a**–**c**), and discarded ones (**d**–**f**).

Although most generated images were realistic enough, i.e., showing at first sight a clear road environment, we had to discard ~82% of the RGB images due to inconsistencies on the road or an unclear intersection. We are currently tackling this issue by introducing new elements to the standard StyleGAN proposal, and we plan to present the outcomes in future work. On the other side, generated warped images showed similar performance when training in a conditional setting, probably due to the amount of information contained in these types of images. The latent space is better regularized, and generated images are closer to the real ones in the embedded space. Therefore, we generate 10,000 warped

images and added them to the warped training set. Figure 15 shows some examples of the generated warped images under a conditional setting. It can be seen that they are realistic when compared to the real ones.



Figure 15. In the first row: some examples of generated images when training the GAN with warped real images. In the second row: examples of real warped images.

5. Experimental Results

The following section presents the extensive experimental results of all the tests we have performed with the different methodologies. These tests have been performed using a NVIDIA DGX-A100 server. The test battery, as described below, has been configured using Weights & Biases [64], which has allowed us to perform an extensive ablation study using the automatic search procedure called *sweep*. By performing a sweep, the Weights & Biases service allows for automatically searching the best combination of hyperparameter values. In order to work with embeddings of similar size, and given that different network architectures have been used, a space reduction should be made in those architectures that return larger feature vectors. To make such an adaptation, a fully connected layer has been used to reduce the size of the embeddings to 512 values. The selected architectures are ResNet and VGG in all its versions, the two versions of MobileNet-V3 and Inception-V3. The test datasets have been divided, regardless of the number of images contained in each dataset, into the following percentages: 70% for the training set, 20% for the validation set, and 10% for the test set. To compare the improvements provided by the proposed methodologies, a sweep has been initially configured to find the best training solution for the proposed networks by direct classification of the RGB and WARPED images. The results obtained can be seen in Table 3, which will be used later in the comparisons. To encourage the research and development in the context of intersection classification as well to allow future researchers to compare their work with respect to ours, our code is available in our institutional repository on https://github.com/invett/nn.based.intersection.classficator, Access Date: 11 September 2021.

			ResNet version				VGG version			MobileNet-V3 version		Inception Version	
			18	34	50	101	152	11	13	16	Large	Small	v 3
	RGB	Validation	0.69	0.71	0.69	0.72	0.70	0.58	0.74	0.73	0.66	0.69	0.65
	1105	Test	-	-	-	-	-	-	0.53	-	-	-	-
KII II-KOAD	Warping	Validation	0.60	0.67	0.70	0.68	0.60	0.60	0.63	0.61	0.65	0.60	0.67
		Test	-	-	0.62	-	-	-	-	-	-	-	-
	RGB	Validation	0.83	0.82	0.77	0.81	0.81	0.80	0.84	0.81	0.82	0.73	0.87
		Test	-	-	-	-	-	-	-	-	-	-	0.78
KII 11-360	TA 7	Validation	0.81	0.78	0.83	0.84	0.80	0.81	0.80	0.82	0.83	0.80	0.87
	warping	Test	-	-	-	-	-	-	-	-	-	-	0.70
	PCB	Validation	0.82	0.87	0.85	0.86	0.80	0.74	0.86	0.85	0.89	0.80	0.89
	KGD	Test	-	-	-	-	-	-	-	-	-	-	0.94
ALCALA	Warning	Validation	0.88	0.91	0.82	0.90	0.88	0.85	0.89	0.85	0.91	0.87	0.91
	warping	Test	-	-	-	-	-	-	-	-	0.92	-	-

Table 3. Baseline results.

Results using direct classification by the proposed architectures using RGB images and bird's eye view images from the three selected datasets. Marked in bold are the best accuracy values for each type of data in both testing and validation.

5.1. Teacher Training

The teacher is trained with artificial images created at run-time. These images try to simulate intersections as they were obtained from a Bird's Eye View, to create an easy yet effective classification problem for the *teacher* network. As the primary use of the *teacher* network is to obtain a set of embeddings that can represent road junctions useful for the training of the *student* network, the loss function used is Triplet loss. As in the previous work, this decision was taken to separate as much as possible the space between the embeddings of different types of intersections so that two similar types are as close as possible and two different types as far apart as possible. Because the images are generated at run-time, once the anchor and the positive samples are chosen, the negative one is selected randomly from the remaining labels. We evaluated the performances of the teacher network using all the proposed architectures as a backbone and a set of 2000 artificial images for training and 1000 for validation. All training runs, regardless of the model, quickly became overfitted. The accuracy values in the training and validation sets reached 100%, so the model stopped learning and was no longer generalized. In case the architecture was too deep, the training was repeated with a customized network, which contained only one or two convolutional blocks (Conv + BN + ReLU) depending on the chosen configuration and a classifier composed of a fully connected layer. The results obtained were similar, so the latter architecture was discarded. In our opinion, this overfitting may be due to the use of the synthetic images created by the crossover generator as those images are effortless to classify. Therefore, the strategy to follow, as we believed that the ease with which the synthetic images were clustered in embeddings was going to be of great help in the training of the student network, was to select as the trained network the checkpoint immediately before the moment in which the precision value in the validation set reached 1. As we can see in Figure 12, performing a clustering analysis over the resulting set of embeddings clearly shows how the CNN can distinguish our intersection classes. As we want to use this information to train the student network, the centroids of each cluster are calculated to be used as reference marks in the subsequent training.

5.2. Student Training

In order to explore the widest solution space possible, a Weights & Biases [64] sweep has been deployed with the following parameters, randomly chosen at each iteration:

- Learning Rate: max: 0.01 min: 2.5×10^{-6}
- Optimizer: adamW, adam, rmsprop, sgd, ASGD, Adamax
- Loss Function: SmoothL1, L1, MSE

• Batch Size: 8, 16, 34, 64, 128

The last two transformations have not been applied to the Alcalá sequences as they do not have stereo or LiDAR information. For more detailed information on datasets and data transformations, see Sections 3 and 4.4.2. In Table 4, we present the results obtained using different input data types (RGB, Warping, 3D-Images and 3D-Masked) for the three used datasets (KITTI-ROAD, KITTI-360, and ALCALÁ). As the training activities have been carried out with Weights & Biases sweeps, the table reports the maximum accuracy values for each architecture. Regarding the testing phases, we reported the value corresponding to the best performance obtained from all the architectures for each input data type. Our experimental activities were addressed to verify:

- 1. first, how the different data transformations exposed in Section 4.4.2 affect the results in the Teacher/Student paradigm and the possible benefits;
- 2. second, whether the use of the Teacher/Student paradigm substantially improves results over the direct classification of the data;
- 3. third, if the data recording methodology and the camera angle are critical points in obtaining good results; and
- 4. fourth, whether the inclusion of new architectures in training produces a significant variation in the results obtained.

If we look at the results in Table 4, we can see no substantial difference in the precision values according to the type of input data. If we take, for example, the results in the KITTI-360 dataset, the lowest validation value is 0.79 and the highest 0.86, a difference of 0.07. The differences in test values are equally minimal, 0.08 between the value achieved with the 3D-Masked data and the value achieved with the RGB images. This pattern seems to be repeated in all the datasets, which leads us to think that the preprocessing of the data does not seem to be such an important issue. If we look at the validation results obtained by the different architectures, there does seem to be a notable difference. For example, in the RGB images of the KITTI-ROAD dataset, there is a 0.25 difference between best and worst architecture. However, although similar differences are repeated between the different datasets and data types, the best- and worst-performing architectures do not seem constant. This variation leads us to think that other factors cause the differences and not the selected architecture. Regarding our second working hypothesis, the results do not seem to be very enlightening. If we compare the results obtained respectively with the baseline and the Teacher/Student approaches (see Tables 3 and 4), the differences between the values achieved on validation are minimal, of the order of 0.01. The accuracy values achieved in testing are also quite similar, the highest difference being 0.11, in the training of the KITTI-ROAD dataset with RGB images. The results obtained in our previous work showed a slight improvement in almost all the Teacher/Student paradigm results concerning the direct classification. However, this improvement does not seem to have appeared during the process of the current research. We believe that this is because using the Teacher/Student paradigm is a risky option, and the results vary considerably depending on the initial configuration of the problem, especially if the data available are not extensive. Finally, to validate our third working hypothesis, we compared the results obtained using the KITTI dataset and both the Baseline and the Teacher/Student approaches (see Tables 3 and 4). As can be seen, the results in the Alcalá dataset, both in direct classification and with the Teacher/Student methodology, are substantially higher than those achieved in KITTI in the same category. A clear example of this is the difference in accuracy between the KITTI-ROAD dataset and the Alcalá dataset when classifying RGB images. In direct classification, the validation value for the Alcalá dataset is one and a half decimal places better than in KITTI-ROAD, and the test value is almost four decimal places higher. These results are repeated within the Teacher/Student paradigm, with the Alcalá dataset being one and a half decimal places better on validation and three decimal places better on testing. These results, in our opinion, are very enlightening. The two KITTI datasets suffer from two things: lack of images and field of view. These two points have been addressed in the Alcalá dataset. We believe that the evident improvement of the results is mainly because the Alcalá dataset has a much

wider field of view than the KITTI datasets. After all, although it also has more images, the difference is not so significant, especially with the dataset KITTI-360 (see Section 3). As the Alcalá dataset is currently a proof of concept, a significant improvement in the image number and field of view will be addressed with a complete dataset that will be available in the next months on http://invett.es/intersectiondataset. In addition, the lack of images in the KITTI dataset also seems to lead to some instability and overfitting in the training that we have observed during Weights & Biases sweeps, see Figure 16. This leads us to think that perhaps the conclusions obtained on our first two hypotheses are not definitive as it would be necessary to work with a much more complete dataset.

			ResNet Version					VGG Version			MobileNet-V3 Version		Inception Version
			18	34	50	101	152	11	13	16	Large	Small	v 3
	DCD	Validation	0.51	0.50	0.53	0.50	0.57	0.66	0.70	0.71	0.75	0.71	0.57
	KGD	Test	-	-	-	-	-	-	-	-	0.64	-	-
	Warning	Validation	0.47	0.58	0.70	0.59	0.52	0.61	0.63	0.66	0.66	0.64	0.57
KITTI ROAD	warping	Test	-	-	0.61	-	-	-	-	-	-	-	-
KITIFKOAD	3D Imagos	Validation	0.45	0.55	0.65	0.67	0.58	0.56	0.62	0.60	0.60	0.62	0.46
	5D-Images	Test	-	-	-	0.64	-	-	-	-	-	-	-
	3D-Masked	Validation	0.64	0.71	0.72	0.71	0.71	0.70	0.72	0.72	0.73	0.72	0.69
		Test	-	-	-	-	-	-	-	-	0.71	-	-
	RGB	Validation	0.60	0.65	0.79	0.79	0.83	0.73	0.82	0.86	0.8	0.79	0.60
		Test	-	-	-	-	-	-	-	0.75	-	-	-
	Warning	Validation	0.63	0.61	0.69	0.8	0.63	0.85	0.79	0.83	0.82	0.73	0.70
KITTI 360	warping	Test	-	-	-	-	-	0.73	-	-	-	-	-
KI111-500	3D-Images	Validation	0.51	0.65	0.71	0.75	0.65	0.61	0.68	0.59	0.79	0.73	0.57
	5D-IIIages	Test	-	-	-	-	-	-	-	-	0.73	-	-
	3D-Masked	Validation	0.65	0.61	0.76	0.78	0.79	0.74	0.77	0.78	0.78	0.76	0.78
	5D-Wasked	Test	-	-	-	-	0.67	-	-	-	-	-	
	DCD	Validation	0.76	0.49	0.88	0.88	0.86	0.82	0.90	0.87	0.88	0.77	0.81
	KGD	Test	-	-	-	-	-	-	0.96	-	-	-	-
ALCALA	Manaina	Validation	0.88	0.87	0.88	0.87	0.91	0.88	0.90	0.80	0.90	0.88	0.90
	warping	Test	-	-	-	-	0.90	-	-	-	-	-	-

Table 4. Teacher/student paradigm results.

Results by dataset and type of data obtained using the Teacher/Student paradigm. Marked in bold are the best accuracy values for each type of data in both testing and validation.

5.3. Metric Learning Training Phase

Due to the results obtained with the Teacher/Student paradigm, and the conclusions we have drawn from them, we believe it is necessary to use the data available to us more extensively to find better results. That idea leads us to the metric learning library [48]. The following process was used in the Teacher/Student methodology to calculate losses when training the student network. First, within each batch, the distance between each student's network embedding and the centroid of the corresponding label is calculated. Then, losses are calculated based on how close the embedding is to the corresponding centroid according to that measure. Finally, losses are balanced according to the weights of each class and averaged. Thus, if the batch has eight data samples, losses will be calculated as a function of eight different distances. Unlike the Teacher/Student methodology, the metric library does not use a teacher to set the reference centroids to each label but uses distance functions to separate/join as much as possible the embeddings returned by the network according to their labeling. This methodology change allows the number of distances computed per batch to be substantially increased as each batch element's distances can be calculated. The actual number of distances to be estimated per batch is given by $\binom{n}{k}$, where *k* is the number of batch elements needed to determine the distance and *n* is the batch size. This exhaustive way of calculating losses during the training process, according to our

criteria, could somehow circumvent the lack of data in the training sets. Our experimental activities were addressed to verify:

- 1. first, whether the new comprehensive way of calculating batch losses helps to alleviate the problems detected in the previous paradigm with the lack of images and
- 2. second, once the lack of images is solved, the usefulness of the data transformations and the improvement over the direct classification can be verified again.

Before discussing the results, we should comment on specific points that differentiate them from those shown in the previous paradigm. As the triplet loss function is implemented within the library and worked quite well in the teacher network training, it is the one that has been chosen to perform all subsequent Weights & Biases training sweeps. In addition, the implementation of this loss function allows selecting between different functions to calculate the distance between embeddings, which allows a broader field of exploration. In order to explore the widest solution space possible, as in the previous methodology, a Weights & Biases [64] sweep has been deployed with the following parameters, randomly chosen at each iteration:

- Learning Rate: max: 0.01 min: 2.5×10^{-6}
- Optimizer: adamW, adam, rmsprop, sgd, ASGD, Adamax
- Distance Function: SNR, Cosine Distance, Pairwise
- Batch Size: 8, 16, 34, 64, 128
- Margin: max: 5.0 min: 0.5 q: 0.5
- Miner: All, Hard



Figure 16. Example of a graph of the training results of the first 100 runs of a Weights & Biases sweep. The plot shows on the ordinates axis the Validation Set, Accuracy metric used to evaluate the performances of our system. As the reader will notice, slightly different parameters lead to considerable differences in the accuracy metric values.

As a comment on the above parameters, the margin is a parameter of the Triplet Margin Loss that can be stated as the desired difference between the anchor-positive distance and the anchor-negative distance. The Metric library offers different options to calculate the model's accuracy from the embeddings returned by the network. During the training, several of them have been registered in order to have a global view of the training, but among all of them, MAPR [51] has been selected to establish which is the best performing model because we believe that it can be the one that best represents a good separation between classes. As we stated in Section 4.2, as MAPR is not reasonably comparable to accuracy and it is necessary to establish a direct classification methodology, the three proposed methodologies have been tested to obtain the results. Because these methodologies have only been used for testing, the validation values of the training cannot be fairly compared with other methodologies such

as Teacher/Student or direct classification. Initially, the training of a fully connected layer to classify the embeddings returned by the architecture was used as a testing methodology. This methodology was discarded since the validation values of the training were relatively low, especially compared to the other methodologies, and therefore the testing values were expected to be even lower. Once this first methodology was discarded, we tested the following two methodologies, SVM and Mahalanobis distance. The results of both methodologies are similar; however, in most cases, the SVM was slightly better, so given the substantially longer computation time required to calculate the covariance matrix to use the Mahalanobis distance, it was decided to use only SVM to perform the tests. Please notice that the values for the Metric Learning paradigm reported in Table 5 always belong to the value returned by this last methodology. If we look at the testing results between Teacher/Student and Metric learning paradigms, see Tables 4 and 5, we can observe that the higher number of comparison examples per batch has not substantially improved accuracy. Many values are similar or differ by a few decimal places, such as the warped images in the KITTI-ROAD dataset or the RGB images in the KITTI-360 dataset. In some cases, as with the 3D images in the KITTI-ROAD dataset, it has even worsened slightly. Therefore, we believe that the same problem of missing images can be seen in both methodologies and that further comparison between the available data has not led to a substantial improvement. In turn, the comparison between the Baseline and the Teacher/Student approach (Tables 3 and 4) yield similar conclusions to those in Section 5.2 because, as mentioned above, the values are still very similar. Focusing on the previous comparisons with the Teacher/Student paradigm, the training using this new methodology seems to validate the assumptions shown in the previous section. If we compare the results obtained in the Alcalá dataset with those obtained in the other two KITTI datasets, we can observe a substantial improvement in accuracy values. The accuracy reaches its highest level, 0.3 higher, compared to the Alcalá dataset and the KITTI-ROAD dataset using RGB images. These results lead us to think that the critical point, as in the previous methodology, is above all the field of view of the camera with which the images are recorded. Figure 17 shows the confusion matrices of the best tests by training methodology and dataset. As can be seen, the results are pretty good, more than it might seem if only the combined accuracy of the seven classes is evaluated. These disaggregated results show that the system's overall performance is good and that with favourable starting conditions, such as a better dataset, it can be excellent.



Figure 17. Different confusion matrix results in (**a**) student KITTI-ROAD, (**b**) student KITTI-360, (**c**) student Alcalá, (**d**) metric KITTI-ROAD, (**e**) metric KITTI-360, and (**f**) metric Alcalá. Please notice that in panels (a,d) we did not introduce straight roads (type-zero intersections).

			ResNet Version				VGG Version			MobileNet-V3 Version		Inception Version	
			18	34	50	101	152	11	13	16	Large	Small	v 3
	DCD	Validation	0.50	0.37	0.52	0.44	0.54	0.41	0.42	0.45	0.48	0.51	0.34
	KGD	Test	-	-	-	-	0.56	-	-	-	-	-	-
	Warning	Validation	0.62	0.47	0.40	0.35	0.46	0.37	0.41	0.35	0.47	0.56	0.38
	warping	Test	0.60	-	-	-	-	-	-	-	-	-	-
KITI-KOAD	3D Imagos	Validation	0.49	0.48	0.41	0.43	0.47	0.29	0.32	0.30	0.44	0.40	0.36
	5D-Images	Test	0.47	-	-	-	-	-	-	-	-	-	-
	3D-Masked	Validation	0.65	0.51	0.51	0.50	0.45	0.46	0.52	0.41	0.49	0.54	0.59
		Test	0.72	-	-	-	-	-	-	-	-	-	-
	RGB	Validation	0.32	0.64	0.64	0.59	0.74	0.59	0.23	0.25	0.55	0.53	0.72
		Test	-	-	-	-	0.69	-	-	-	-	-	-
	Warning	Validation	0.63	0.65	0.31	0.42	0.38	0.65	0.31	0.60	0.64	0.41	0.44
KITTI 360	warping	Test	-	0.78	-	-	-	-	-	-	-	-	-
KI111-300	3D Imagos	Validation	0.57	0.53	0.29	0.40	0.62	0.36	0.42	0.33	0.67	0.56	0.60
	5D-mages	Test	-	-	-	-	-	-	-	-	0.73	-	-
	3D-Masked	Validation	0.65	0.61	0.76	0.78	0.79	0.74	0.77	0.78	0.78	0.76	0.78
	5D-Wasked	Test	-	-	-	-	0.81	-	-	-	-	-	
	DCD	Validation	0.72	0.78	0.83	0.75	0.50	0.80	0.45	0.84	0.70	0.60	0.40
	KGD	Test	-	-	-	-	-	-	-	0.94	-	-	-
ALCALA	Manning	Validation	0.74	0.81	0.72	0.52	0.81	0.71	0.77	0.59	0.84	0.83	0.82
	vvarping	Test	-	-	-	-	-	-	-	-	0.91	-	-

Table 5. Metric learning paradigm results.

Results by dataset and type of data obtained using the metric learning paradigm. For each type of data, we highlighted in bold the best MAPR for validation and the best accuracy for testing.

5.4. Multi-Frame Scheme Results

It seems a reasonably straightforward deduction that vehicle traffic is an event that runs over time. The approach to an intersection is not an event that only exists at point t. As the vehicle progresses to the junction, it seems logical to think it will receive more information. Therefore, it appears that if we can group all the information from instant t to instant t + n when the car is already at the intersection, we should achieve greater accuracy when classifying each intersection. With this idea in mind, the next step we have considered is to group, in order of approach, all the embeddings that belong to the same intersection to be classified as a *single sequence*. Our experimental activities were addressed to verify

1. first, we check if the temporal integration helps achieving our goal of classifying intersections and

2. second evaluate whether this new approach offers more light on the previous results.

As explained in Section 4.5.1, the inclusion of an RNN was the most straightforward methodology to implement as it can directly work with embeddings and integrate them. Training has been performed with two selected RNN architectures: GRU and LSTM. Preliminary results for both structures were very similar with those belonging to the LSTM network being slightly superior. Therefore, the subsequent training was performed with a LSTM architecture. Tests have also been performed in such a way that it is possible to obtain a final classification of the embeddings returned by the RNN, being the options to choose, as specified in Section 4.5.1, SVM, Mahalanobis, and Fully connected. In this case, unlike in the single-frame approaches, the best results have been obtained by training a fully connected layer while training the RNN network architecture. In order to explore the most comprehensive solution space possible, as in the single-frame approaches, a Weights & Biases [64] sweep has been deployed with the following parameters, randomly chosen at each iteration:

- Learning Rate: max: 0.01 min: 2.5×10^{-6}
- Optimizer: adamW, adam, rmsprop, sgd, ASGD, Adamax
- Loss Function: Cross Entropy, Focal
- Batch Size: 16, 34, 64
- FC dropout: max: 0.5 min: 0.1 q: 0.1
- LSTM dropout: max: 0.5 min: 0.1 q: 0.1
- LSTM hidden layer: 256, 128, 64, 32, 16, 8
- LSTM layers: 1, 2

As a comment to the above parameters, the dropout of the fully connected layer is located just between the embedding and the classifier. The Focal loss function has been implemented from [65]. Regarding our first assumption, looking at Table 6, it does not seem that temporal integration has significantly changed the results. The validation values have increased in some cases, as in the KITTI-ROAD 3D images, but the test value is significantly worse, indicating overfitting. However, if we look at the results in the Alcalá dataset, the values for the two available paradigms are relatively similar. This connection leads us to think that the problems from previous approaches have been transferred to this one. Some comparisons were made between the results obtained by the LSTM and the results obtained by the single-frame approach. A simple temporal integration was made by a voting process between the results of all the frames of the same sequence to make a fair analysis. The results were very similar and analyzing them in-depth leads us to confirm the assumptions in previous points. The problem lies on two fronts: the number of images and the viewing angle. The lack of images means that when grouping them in sequences, the examples used to train the LSTM are smaller than in a single-frame approach, so it seems logical that the overfitting has increased. The lack of viewing angle in the KITTI datasets means that the network does not have enough information to extract good features from the crossing, something that cannot be solved with temporal information. If the feature extractor cannot divide the embeddings correctly, the LSTM will not be able to classify accurately.

	KITTI RO	DAD	KITTI-3	60	Alcalá		
	Validation	Test	Validation	Test	Validation	Test	
RGB	0.69	0.55	0.90	0.85	0.85	0.84	
Warping	0.82	0.45	0.82	0.73	0.91	0.92	
3D-Images	0.88	0.65	0.84	0.79	-	-	
3D-Masked	0.90	0.70	0.89	0.81	-	-	

Table 6. Multi-frame paradigm results.

Results by dataset and type of data obtained using temporal integration schemes. As the reader can see, the accuracies in testing greatly decrease in all configurations on KITTI and KITTI-360 datasets, but not with the Alcalá dataset. Finally, please notice that Alcalá dataset does not have 3D-Images and 3D-Masked values, as no stereo configuration and LiDAR data are available.

5.5. GAN Related Results

We generated 10,000 RGB images and 10,000 warped images. As stated in Section 4.6, we use a conditional setting for the warped images but a normal one for the RGB images, as we noticed an important detriment in performance. Figures 14 and 15 show some examples of RGB and warped generated images, respectively. Before including the artificially generated images in our original train dataset, we tested if they were close to the real ones by feeding them to our model trained with images of KITTI-ROAD, KITTI-360, and our datasets, and comparing the codes in the embedding space. In order to do so, we calculated the distances of these codes to the nearest clusters' centroids previously computed with the real images. We did the same with a random dataset and checked the statistics of the three setups, showing consistency among the generated and real images, while random images' embeddings were far from all centroids. Table 7 shows accuracy results for the Teacher/Student paradigm as well as for metric learning scheme for both RGB and warped

images. Please note that we use MAP metric for validation results in the metric learning scheme. Dataset *ALL* refers to KITTI-ROAD, KITTI-360, Alcalá-1, and Alcalá-2, while *ALL AUGMENTED* refers to the same datasets together with the generated images. An improvement in test performance can be seen for the four setups when augmenting the dataset with GAN images. In the RGB setting, validation accuracy is lower in the case of the augmented dataset, which reaffirms our hypothesis that it improves generalizability.

		War	ped		RGB					
	Teacher/Student		Metric Lea	rning	Teacher/St	udent	Metric Learning			
	Validation	Test	Validation	Test	Validation	Test	Validation	Test		
ALL	0.890	0.894	0.901	0.872	0.885	0.894	0.893	0.817		
ALL AUGMENTED (GAN)	0.932	0.935	0.920	0.928	0.876	0.935	0.882	0.935		

Table 7. Results with GAN-Augmented Dataset.

Marked in bold are the best accuracy value for each type of data and each training methodology in both testing and validation.

5.6. Pytorchvideo

Our pilot experiments have been performed using ResNet3D and X3D [58] networks, together with the KITTI-360 and Alcalá datasets. Few changes to the original code were applied. Among them, we removed all the mirroring data augmentation schemes in the Charades dataloader implementation and modified the number of classes from 157 to our 7 basic geometries. Another distinctive feature of our experiments is related to the length of the input data, which strongly differs from the original video feed. As the intersections' frames were manually selected, we modified the part of the code where a clip is randomly sampled from a whole video sequence. Our experimental activities were addressed to verify the following:

- First, how the video-analysis networks handle both RGB and MBEVs images, comparing their performances with respect to the learning schemes proposed in this research. This test was executed using the KITTI360 and KITTI360-masked imagery.
- 2. Second, how different frame rates affect the classification capabilities. For this second test, we used the Alcalá-1 sequence, subsampling the original 30 fps to 6 and 15 fps. These experiments could not have been performed with the KITTI sequences due to the dataset low frame rate. Please notice that all the *actions* datasets originally used with these networks share the same length, which is not realistic with videos containing intersection approaches due to different speeds of the vehicle while approaching a generic intersection.

We repeated each of these tests using both the aforementioned ResNet3D and X3D approaches. We trained the system on a NVIDIA DGX-A100 server, following the video classification tutorial provided in the PytorchVideo GitHub repository, a variable epoch number between 200 and 1000 iterations with the CosineAnnealingLR Pytorch scheduler and the AdamW optimizer with different learning rates between 1×10^{-1} and 1×10^{-5} and weight decay set to 1×10^{-4} . As for the image size, similarly to all the other approaches, we used a standard 224×244 size. Regarding the first experiment, tests showed us that different clip frame rates do not have a significant impact on the prediction performances, see Figure 18. This is of particular interest and suggests that high frame rates are not strictly required for this analysis, suggesting that the network can infer the geometry with just a few frames. A first interesting comparison can be made with respect to the different frame length experiment. As shown in Figure 19c, the performances of the LSTM system are better than all the different experiments performed with both the ResNet3D and the X3D models and the different time lengths used. Note that the Pytorch implementation of the LSTM considers different frame lengths, something that by its nature PytorchVideo framework does not. This is a clear advantage in favor of LSTMs approaches, as the

approach to an intersection trivially depends on the vehicle's speed. This issue might be solved with approaches that involve multiple pathways for activity recognition, such as the SlowFast [66] approach. However, due to technical issues of the PytorchVideo code, which is nowadays in its infancy, we were unable to further investigate this model. Last, given the results, we can conclude that the investigated networks do not achieve better results with respect to the RGB inputs. The second experiment further corroborates the latter opinion, and here the advantages of MBEV images are undisputed. As shown in Figure 19a–d, both the transformed images have better results with respect to the corresponding RGB images. Regarding the comparison between ResNet3D and X3D architectures, no clear advantage within the two architectures has been observed. The comparison with respect to the most similar temporal sequence-oriented architectures is shown in Figure 19. As the reader can see from Figure 19a, the LSTM approach outperforms both ResNet3D and X3D approaches of Figure 19a, while the corresponding MBEV experiments obtain much more similar results, see Figure 19b vs. Figure 20b,d, respectively, for ResNet3D and X3D.



Figure 18. Different frame length test. (**a**,**b**) ResNet3D model with input at 15 or 6 fps. (**c**,**d**) X3D model with input at 15 or 6 fps. Dataset: Alcalá-1, test-split, 51 total intersections.



Figure 19. LSTM results using KITTI360 dataset and RGB data (**a**), MBEVs (**b**), and Alcalá-1 RGB dataset (**c**).



Figure 20. Different view test. (**a**,**b**): ResNet3D model, using RGB and MBEVs, respectively. (**c**,**d**): X3D model, using RGB and MBEVs, respectively. Dataset: KITTI360, test-split, 48 total intersections.

6. Conclusions

This work presented an in-depth study of the intersection classification problem. Toward this goal, we explored the capabilities of state-of-the-art neural network approaches for image classification, considering different approaches such as single-frame, videoanalysis and 3D-networks, non-traditional learning techniques for the specific intersection analysis subject, as well as different data source, e.g., RGB cameras and LiDARs sensors. An extensive experimental activity has been performed to validate each considered approach and its contribution towards the intersection classification problem. As a conclusion to this research work, we can obtain the following statements. First, as also mentioned in [21], is the critical importance of the dataset. As seen throughout the presented results, the number of available intersections plays a crucial role with all the tested approaches, achieving better results with the dataset with more intersection sequences. Despite the assertion's simplicity, we claim that a complete analysis of the issue still requires a much broader set of intersections. Past state-of-the-art approaches limited to ad hoc datasets or the KITTI dataset series represent an essential yet limited contribution to the whole problem. For this reason we plan to release our recorded intersection dataset together with the *per-frame* and *per-sequence* annotations. Our contribution will provide a meta dataset containing not only our multi-season sequences but also meaningful annotations of current state-of-the-art datasets. Nevertheless, even with the aforementioned dataset at its present state, this research allowed us to prove functional requirements for the scope. Among these, the angle of view of the camera with which it was recorded is critical. Results show that detecting the intersection needs as much information as possible about what stands in front of the vehicle. In such a case, with limited environmental information from the imagery, e.g., with the KITTI imagery as discussed in Section 5.4 and Table 6, simply integrating temporal data in the form of the output of CNN networks can frustrate all the efforts and benefits offered by RNN variants. On the other hand, the results do not seem conclusive when using other training paradigms such as Teacher/Student or metric learning, or temporal integration. The results show no significant difference between these paradigms and direct classification. However, based on our experience, we believe that this is due to the limitations of the datasets. Therefore, it is possible that if these tests were repeated on a larger dataset, the results could be much more decisive than those obtained during this research. Same conclusions can be drawn from the preprocessing of the data, e.g., Bird's Eye View. Currently, there is no conclusive evidence that image transformations represent a benefit or a detriment towards the intersection classification process. Regarding the scarcity of training and testing data, this work presented two different contributions. First, we presented a preliminary dataset of intersections recorded in the surrounding area of the Universidad de Alcalá. This cross-season dataset will be published in the following months, and we consider it will set an important milestone in the context of the intersection classification. Second, we prepared a new artificial set of intersections implemented using a Generative Adversarial Network approach which showed an overall improvement in the results. To the best of our knowledge, this was the first time that this approach has been tested in the context of intersection classification. We believe that combining authentic images with artificial images would be ideal, given the ease of obtaining the latter and as a means to balance the dataset. This second research line is of great interest, and we plan to write a second paper focusing only on this line of work. First, we want to explore the regularization of the latent space in order to achieve controllability in the generation process beyond the type of intersection. This would allow us to gain insights about the inner workings of our classification network and shed some light on which are the most essential features for achieving good performance. Second, by introducing our network in the GAN training loop, we expect to achieve more realistic intersection images, where the generative network focuses on what is most important for our problem.

Author Contributions: Conceptualization, A.L.B. and Á.H.S.; methodology, A.L.B. and Á.H.S.; software, A.L.B., Á.H.S., S.C.L. and J.L.; validation, A.L.B. and Á.H.S.; formal analysis, A.L.B. and Á.H.S.; investigation, A.L.B. and Á.H.S.; resources, I.P.A. and M.Á.S.; data curation, A.L.B. and Á.H.S.; writing—original draft preparation, A.L.B. and Á.H.S. and S.C.L. and J.L.; writing—review and editing, A.L.B. and Á.H.S. and S.C.L. and J.L. and N.H.P. and I.G.D.; supervision, A.L.B. and M.Á.S.; project administration, A.L.B. and I.P.A. and M.Á.S.; funding acquisition, I.P.A. and M.Á.S. All authors have read and agreed to the published version of the manuscript.

Funding: The work of Augusto Luis Ballardini has been funded by European Union H2020, under Grant Agreement Marie Skłodowska-Curie n. 754382 Got Energy Talent (GET). The content of this publication does not reflect the official opinion of the European Union. Responsibility for the information and views expressed herein lies entirely with the author. This work was also supported in part by Spanish Ministry of Science and Innovation (Research Grants DPI2017-90035-R and PID2020-114924RB-I00) and in part by the Community Region of Madrid (Research Grant 2018/EMT-4362 SEGVAUTO 4.0-CM).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Regarding KITTI and KITTI360, the data presented in this study is openly available on http://www.cvlibs.net/datasets/kitti doi: 10.1177/0278364913491297 Access Date: 11 September 2021, and http://www.cvlibs.net/datasets/kitti-360 doi: 10.1109/CVPR.2016.401 Access Date: 11 September 2021; the Alcalá-1/Alcalá-2 datasets will be published on http://invett. es/intersectiondataset.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- 1. European Commission Road Safety Key Figures 2020. Available online: https://ec.europa.eu/transport/road_safety/sites/ roadsafety/files/pdf/scoreboard_2020.pdf (accessed on 21 May 2021).
- European Union Annual Accident Report 2018. Available online: https://ec.europa.eu/transport/road_safety/specialist/ observatory/statistics/annual_accident_report_archive_en (accessed on 21 May 2021).
- 3. Fatality and Injury Reporting System Tool of U.S. National Highway Traffic Safety Administration. Available online: https://cdan.dot.gov/query (accessed on 21 May 2021).
- 4. European Union Report on Mobility and Transportation: ITS & Vulnerable Road Users. Available online: https://ec.europa.eu/ transport/themes/its/road/action_plan/its_and_vulnerable_road_users_en (accessed on 3 September 2021).
- Huang, X.; Cheng, X.; Geng, Q.; Cao, B.; Zhou, D.; Wang, P.; Lin, Y.; Yang, R. The ApolloScape Dataset for Autonomous Driving. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition (CVPR) Workshops, Salt Lake City, UT, USA, 18–22 June 2018.
- Chang, M.F.; Lambert, J.W.; Sangkloy, P.; Singh, J.; Bak, S.; Hartnett, A.; Wang, D.; Carr, P.; Lucey, S.; Ramanan, D.; et al. Argoverse: 3D Tracking and Forecasting with Rich Maps. In Proceedings of the Conference on Computer Vision and Pattern Recognition (CVPR), Long Beach, CA, USA, 16–20 June 2019.
- Houston, J.; Zuidhof, G.; Bergamini, L.; Ye, Y.; Jain, A.; Omari, S.; Iglovikov, V.; Ondruska, P. One Thousand and One Hours: Self-driving Motion Prediction Dataset. *arXiv* 2020, arXiv:2006.14480.
- 8. Caesar, H.; Bankiti, V.; Lang, A.H.; Vora, S.; Liong, V.E.; Xu, Q.; Krishnan, A.; Pan, Y.; Baldan, G.; Beijbom, O. nuScenes: A multimodal dataset for autonomous driving. *arXiv* 2019, arXiv:1903.11027.
- 9. PandaSet (2020) PandaSet Dataset. Available online: https://scale.com/open-datasets/pandaset (accessed on 21 May 2021).
- 10. Geiger, A.; Lenz, P.; Stiller, C.; Urtasun, R. Vision meets Robotics: The KITTI Dataset. Int. J. Robot. Res. (IJRR) 2013, 32, 1231–1237. [CrossRef]
- 11. Xie, J.; Kiefel, M.; Sun, M.T.; Geiger, A. Semantic Instance Annotation of Street Scenes by 3D to 2D Label Transfer. In Proceedings of the Conference on Computer Vision and Pattern Recognition (CVPR), Las Vegas, NV, USA, 27–30 June 2016.
- 12. Goodfellow, I.J.; Pouget-Abadie, J.; Mirza, M.; Xu, B.; Warde-Farley, D.; Ozair, S.; Courville, A.; Bengio, Y. Generative Adversarial Networks. *arXiv* 2014, arXiv: 1406.2661.
- 13. Ki, Y.; Lee, D. A Traffic Accident Recording and Reporting Model at Intersections. *IEEE Trans. Intell. Transp. Syst.* 2007, *8*, 188–194. [CrossRef]
- 14. Golembiewski, G.; Chandler, B.E. *Intersection Safety: A Manual for Local Rural Road Owners*; Technical Report; United States Federal Highway Administration Office of Safety: Washington, DC, USA, 2011.

- 15. Kushner, T.R.; Puri, S. Progress in road intersection detection for autonomous vehicle navigation. In *Mobile Robots II. International Society for Optics and Photonics*; SPIE: Cambridge, CA, USA, 1987; Volume 852, pp. 19–24.
- 16. Geiger, A. *Probabilistic Models for 3D Urban Scene Understanding from Movable Platforms;* KIT Scientific Publishing: Karlsruhe, Germany, 2013; Volume 25.
- 17. An, J.; Choi, B.; Sim, K.B.; Kim, E. Novel intersection type recognition for autonomous vehicles using a multi-layer laser scanner. *Sensors* **2016**, *16*, 1123. [CrossRef]
- 18. Ballardini, A.L.; Cattaneo, D.; Fontana, S.; Sorrenti, D.G. An online probabilistic road intersection detector. In Proceedings of the IEEE International Conference on Robotics and Automation (ICRA), Singapore, 29 May–3 June 2017.
- Habermann, D.; Vido, C.E.; Osório, F.S.; Ramos, F. Road junction detection from 3d point clouds. In Proceedings of the 2016 International Joint Conference on Neural Networks (IJCNN), Vancouver, BC, Canada, 24–29 July 2016; pp. 4934–4940.
- Baumann, U.; Huang, Y.Y.; Gläser, C.; Herman, M.; Banzhaf, H.; Zöllner, J.M. Classifying road intersections using transfer-learning on a deep neural network. In Proceedings of the 2018 21st International Conference on Intelligent Transportation Systems (ITSC), Maui, HI, USA, 4–7 November 2018; pp. 683–690.
- Bhatt, D.; Sodhi, D.; Pal, A.; Balasubramanian, V.; Krishna, M. Have i reached the intersection: A deep learning-based approach for intersection detection from monocular cameras. In Proceedings of the 2017 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), Vancouver, BC, Canada, 24–28 September 2017; pp. 4495–4500. [CrossRef]
- 22. Hochreiter, S.; Schmidhuber, J. Long Short-term Memory. Neural Comput. 1997, 9, 1735–1780. [CrossRef] [PubMed]
- Koji, T.; Kanji, T. Deep intersection classification using first and third person views. In Proceedings of the 2019 IEEE Intelligent Vehicles Symposium (IV), Paris, France, 9–12 June 2019; pp. 454–459.
- 24. Xu, P.; Davoine, F.; Bordes, J.B.; Zhao, H.; Denœux, T. Multimodal information fusion for urban scene understanding. *Mach. Vis. Appl.* **2016**, *27*, 331–349. [CrossRef]
- 25. Vinyals, O.; Toshev, A.; Bengio, S.; Erhan, D. Show and Tell: A Neural Image Caption Generator. arXiv 2015, arXiv:1411.4555.
- 26. Donahue, J.; Hendricks, L.A.; Rohrbach, M.; Venugopalan, S.; Guadarrama, S.; Saenko, K.; Darrell, T. Long-term Recurrent Convolutional Networks for Visual Recognition and Description. *arXiv* **2016**, arXiv:1411.4389.
- 27. Ng, J.Y.H.; Hausknecht, M.; Vijayanarasimhan, S.; Vinyals, O.; Monga, R.; Toderici, G. Beyond Short Snippets: Deep Networks for Video Classification. *arXiv* 2015, arXiv:1503.08909.
- Shi, X.; Chen, Z.; Wang, H.; Yeung, D.Y.; Wong, W.; Woo, W. Convolutional LSTM Network: A Machine Learning Approach for Precipitation Nowcasting. *arXiv* 2015, arXiv:1506.04214.
- 29. Sandfort, V.; Yan, K.; Pickhardt, P.; Summers, R. Data augmentation using generative adversarial networks (CycleGAN) to improve generalizability in CT segmentation tasks. *Sci. Rep.* **2019**, *9*, 16884. [CrossRef] [PubMed]
- Frid-Adar, M.; Diamant, I.; Klang, E.; Amitai, M.; Goldberger, J.; Greenspan, H. GAN-based synthetic medical image augmentation for increased CNN performance in liver lesion classification. *Neurocomputing* 2018, 321, 321–331. [CrossRef]
- 31. Perez, L.; Wang, J. The Effectiveness of Data Augmentation in Image Classification using Deep Learning. *arXiv* 2017, arXiv:1712.04621.
- 32. Antoniou, A.; Storkey, A.; Edwards, H. Data Augmentation Generative Adversarial Networks. arXiv 2018, arXiv:1711.04340.
- 33. Wu, E.; Wu, K.; Cox, D.; Lotter, W. Conditional Infilling GANs for Data Augmentation in Mammogram Classification. In *Image Analysis for Moving Organ, Breast, and Thoracic Images*; Stoyanov, D., Taylor, Z., Kainz, B., Maicas, G., Beichel, R.R., Martel, A., Maier-Hein, L., Bhatia, K., Vercauteren, T., Oktay, O., et al., Eds.; Springer International Publishing: Cham, Switzerland, 2018.
- 34. Bowles, C.; Gunn, R.; Hammers, A.; Rueckert, D. GANsfer Learning: Combining labelled and unlabelled data for GAN based data augmentation. *arXiv* **2018**, arXiv:1811.10669.
- 35. Li, C.L.; Zaheer, M.; Zhang, Y.; Poczos, B.; Salakhutdinov, R. Point Cloud GAN. arXiv 2018, arXiv:1810.05795.
- 36. Gadelha, M.; Rai, A.; Maji, S.; Wang, R. Inferring 3D Shapes from Image Collections using Adversarial Networks. *arXiv* 2019, arXiv:1906.04910.
- 37. Saito, M.; Matsumoto, E.; Saito, S. Temporal Generative Adversarial Nets with Singular Value Clipping. *arXiv* 2017, arXiv:1611.06624.
- 38. Karras, T.; Laine, S.; Aila, T. A Style-Based Generator Architecture for Generative Adversarial Networks. *arXiv* 2019, arXiv:1812.04948.
- 39. Karras, T.; Laine, S.; Aittala, M.; Hellsten, J.; Lehtinen, J.; Aila, T. Analyzing and Improving the Image Quality of StyleGAN. *arXiv* **2020**, arXiv:1912.04958.
- 40. Gal, R.; Cohen, D.; Bermano, A.; Cohen-Or, D. SWAGAN: A Style-based Wavelet-driven Generative Model. *arXiv* 2021, arXiv:2102.06108.
- 41. Karras, T.; Aittala, M.; Hellsten, J.; Laine, S.; Lehtinen, J.; Aila, T. Training Generative Adversarial Networks with Limited Data. *arXiv* 2020, arXiv:2006.06676.
- 42. Ballardini, A.L.; Hernández, Á.; Sotelo, M.Á. Model Guided Road Intersection Classification. arXiv 2021, arXiv:2104.12417.
- 43. He, K.; Zhang, X.; Ren, S.; Sun, J. Deep Residual Learning for Image Recognition. In Proceedings of the IEEE Computer Vision and Pattern Recognition (CVPR), Las Vegas, NV, USA, 27–30 June 2016.
- 44. Simonyan, K.; Zisserman, A. Very deep convolutional networks for large-scale image recognition. arXiv 2014, arXiv:1409.1556.

- Howard, A.; Sandler, M.; Chu, G.; Chen, L.C.; Chen, B.; Tan, M.; Wang, W.; Zhu, Y.; Pang, R.; Vasudevan, V.; et al. Searching for mobilenetv3. In Proceedings of the IEEE/CVF International Conference on Computer Vision, Seoul, Korea, 27–28 October 2019; pp. 1314–1324.
- 46. Szegedy, C.; Vanhoucke, V.; Ioffe, S.; Shlens, J.; Wojna, Z. Rethinking the inception architecture for computer vision. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Las Vegas, NV, USA, 27–30 June 2016; pp. 2818–2826.
- 47. Bellet, A.; Habrard, A.; Sebban, M. A Survey on Metric Learning for Feature Vectors and Structured Data. *arXiv* 2014, arXiv:1306.6709.
- 48. Musgrave, K.; Belongie, S.; Lim, S.N. PyTorch Metric Learning. arXiv 2020, arXiv:2008.09164.
- Yuan, T.; Deng, W.; Tang, J.; Tang, Y.; Chen, B. Signal-to-noise ratio: A robust distance metric for deep metric learning. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Long Beach, CA, USA, 16–20 June 2019; pp. 4815–4824.
- Schroff, F.; Kalenichenko, D.; Philbin, J. FaceNet: A unified embedding for face recognition and clustering. In Proceedings of the 2015 IEEE Conference on Computer Vision and Pattern Recognition (CVPR), Boston, MA, USA, 7–12 June 2015; pp. 815–823. [CrossRef]
- 51. Musgrave, K.; Belongie, S.; Lim, S.N. A metric learning reality check. In *European Conference on Computer Vision*; Springer: Berlin/Heidelberg, Germany, 2020; pp. 681–699.
- 52. Ballardini, A.L.; Cattaneo, D.; Sorrenti, D.G. Visual Localization at Intersections with Digital Maps. In Proceedings of the IEEE International Conference on Robotics and Automation (ICRA), Montreal, QC, Canada, 20–24 May 2019.
- 53. Xu, H.; Zhang, J. AANet: Adaptive Aggregation Network for Efficient Stereo Matching. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Seattle, WA, USA, 14–19 June 2020; pp. 1959–1968.
- Hernández, A.; Woo, S.; Corrales, H.; Parra, I.; Kim, E.; Llorca, D.F.; Sotelo, M.A. 3D-DEEP: 3-Dimensional Deep-learning based on elevation patterns for road scene interpretation. In Proceedings of the 2020 IEEE Intelligent Vehicles Symposium (IV), Las Vegas, NV, USA, 19 October–13 November 2020; pp. 892–898. [CrossRef]
- 55. Cattaneo, D.; Vaghi, M.; Fontana, S.; Ballardini, A.L.; Sorrenti, D.G. Global visual localization in LiDAR-maps through shared 2D-3D embedding space. In Proceedings of the 2020 IEEE International Conference on Robotics and Automation (ICRA), Philadelphia, PA, USA, 23–27 May 2020; pp. 4365–4371. [CrossRef]
- 56. Paszke, A.; Gross, S.; Chintala, S.; Chanan, G.; Yang, E.; DeVito, Z.; Lin, Z.; Desmaison, A.; Antiga, L.; Lerer, A. Automatic differentiation in PyTorch. Advances in Neural Information Processing Systems, Autodiff Workshop. 2017. Available online: https://openreview.net/forum?id=BJJsrmfCZ (accessed on 21 May 2021).
- 57. Carreira, J.; Zisserman, A. Quo vadis, action recognition? A new model and the kinetics dataset. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Honolulu, HI, USA, 21–26 July 2017; pp. 6299–6308.
- Feichtenhofer, C. X3d: Expanding architectures for efficient video recognition. In Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, Seattle, WA, USA, 14–19 June 2020; pp. 203–213.
- 59. Kay, W.; Carreira, J.; Simonyan, K.; Zhang, B.; Hillier, C.; Vijayanarasimhan, S.; Viola, F.; Green, T.; Back, T.; Natsev, P.; et al. The kinetics human action video dataset. *arXiv* 2017, arXiv:1705.06950.
- 60. Sigurdsson, G.A.; Varol, G.; Wang, X.; Farhadi, A.; Laptev, I.; Gupta, A. Hollywood in Homes: Crowdsourcing Data Collection for Activity Understanding. *arXiv* 2016, arXiv:1604.01753.
- 61. Damen, D.; Doughty, H.; Farinella, G.M.; Fidler, S.; Furnari, A.; Kazakos, E.; Moltisanti, D.; Munro, J.; Perrett, T.; Price, W.; et al. The EPIC-KITCHENS Dataset: Collection, Challenges and Baselines. *IEEE Trans. Pattern Anal. Mach. Intell. (TPAMI)* **2020**. [CrossRef] [PubMed]
- Goyal, R.; Kahou, S.E.; Michalski, V.; Materzynska, J.; Westphal, S.; Kim, H.; Haenel, V.; Fruend, I.; Yianilos, P.; Mueller-Freitag, M.; et al. The "Something Something" Video Database for Learning and Evaluating Visual Common Sense. In Proceedings of the 2017 IEEE International Conference on Computer Vision (ICCV), Venice, Italy, 22–29 October 2017; pp. 5843–5851. [CrossRef]
- 63. Fan, H.; Li, Y.; Xiong, B.; Lo, W.Y.; Feichtenhofer, C. PySlowFast. 2020. Available online: https://github.com/facebookresearch/slowfast (accessed on 21 May 2021).
- 64. Biewald, L. Experiment Tracking with Weights and Biases. 2020. Software. Available online: wandb.com (accessed on 21 May 2021).
- 65. Lin, T.Y.; Goyal, P.; Girshick, R.; He, K.; Dollár, P. Focal loss for dense object detection. In Proceedings of the IEEE International Conference on Computer Vision, Venice, Italy, 22–29 October 2017; pp. 2980–2988.
- 66. Feichtenhofer, C.; Fan, H.; Malik, J.; He, K. SlowFast Networks for Video Recognition. In Proceedings of the IEEE/CVF International Conference on Computer Vision (ICCV), Seoul, Korea, 27–28 October 2019.